Author's response to reviews

Title: Understanding the Management of Electronic Test Result Notifications in the Outpatient Setting

Authors:

  Sylvia J Hysong (sylvia.hysong@va.gov)
  Mona K Sawhney (mona.sawhney@va.gov)
  Lindsey A Wilson (Lindsey.aw@gmail.com)
  Dean F Sittig (dean.f.sittig@uth.tmc.edu)
  Adol Esquivel (adol@bioinformaticsinmedicine.org)
  Simran Singh (simran.singh@va.gov)
  Hardeep Singh (hardeep.singh@va.gov)

Version: 2 Date: 11 March 2011

Author's response to reviews: see over
March 11, 2011

Jigisha Patel MRCP, PhD, Series Editor
BMC Medical Informatics and Decision Making

Dear Dr. Patel:

Thank you very much for the many insightful comments you and the reviewers provided us regarding our manuscript # 1622472935510429 - Reducing Missed Test Results in Electronic Health Records: Thinking Beyond Technology”. We have carefully considered each of the comments and have incorporated them in the manuscript. Please note that we have changed the title of the paper to Understanding the Management of Electronic Test Result Notifications in the Outpatient Setting, in direct response to reviewer requests.

Following is a detailed account of our responses to each of the reviewers’ comments which we hope will be found satisfactory. We look forward to seeing our manuscript in print.

If you have additional questions about the changes to the manuscript, or if I can be of any further assistance, please do not hesitate to contact me.

Sincerely,

Sylvia J. Hysong, Ph.D.
Assistant Professor

Backup contact:
Hardeep Singh, MD, MPH
Hardeeps@bcm.edu